Author’s response to reviews

Title: Patient safety in Dutch primary care: a study protocol

Authors:

Mirjam Harmsen (M.Harmsen@iq.umcn.nl)
Sander Gaal (S.Gaal@iq.umcn.nl)
Simone van Dulmen (S.vanDulmen@iq.umcn.nl)
Eimert de Feijter (E.deFeijter@iq.umcn.nl)
Paul Giesen (P.Giesen@iq.umcn.nl)
Annelies Jacobs (J.Jacobs@iq.umcn.nl)
Lucie Martijn (L.Martijn@iq.umcn.nl)
Theodorus Mettes (D.Mettes@dent.umcn.nl)
Wim Verstappen (W.Verstappen@iq.umcn.nl)
Ria Nijhuis-van der Sanden (R.Nijhuis@iq.umcn.nl)
Michel Wensing (M.Wensing@iq.umcn.nl)

Version: 7 Date: 24 June 2010

Author’s response to reviews: see over
Cover letter to paper 1963107263320037 (June 24, 2010)

Mirjam Harmsen and co-authors

Major revisions

Tables:
- We have adjusted the tables, so they are divided into rows and columns, and do not include vertical lines. We used the MS Word auto-format template Simple 1.
- We have removed Box 1 (Definition of a patient safety incident) from the paper, since we already included the information in the text and the box did not add information.
- We have removed the bullet points.
- We have removed the shading and colours within the tables.

Minor revisions

- ‘Coached study patients’ on page 6 was changed into ‘coached patients’.
- We have corrected some typographic errors.